Author’s response to reviews

Title: Robotic assisted-bronchoscopy: technical tips and lessons learned from the initial experience with sampling peripheral lung lesions

Authors:

Septimiu Murgu (smurgu@medicine.bsd.uchicago.edu)

Version: 2 Date: 23 Apr 2019

Author’s response to reviews:

Dear Editors,

Please find attached the revised manuscript entitled “Robotic assisted-bronchoscopy: technical tips and lessons learned from the initial experience with sampling peripheral lung lesions”.

As requested, I formatted the abstract to include the following subheadings: Background, Methods, Results, Conclusions. I also ensured that all headings within the main manuscript adhere to the submission guidelines. The Keywords section and Consent for publication statement have now added, as per the journal’s guidelines. In addition, the figures in a higher resolution format have been re-uploaded.

Sincerely,

SD Murgu